Author’s response to reviews

Title: A European study on alcohol and drug use among young drivers: the TEND by Night study design and methodology.

Authors:

Roberta Siliquini (roberta.siliquini@unito.it)
Simone Chiadò Piat (simone.chiado@unito.it)
Francisco Alonso (francisco.alonso@uv.es)
Axel Druart (axel@ryd.nl)
Marcin Kedzia (mkedzia@holowczyc.pl)
Antonio Mollica (guidasicura@motoroasi.com)
Valeria Siliquini (v.siliquini@setinweb.it)
Daniel Vankov (chairman@open-youth.org)
Anita Villerusa (anita.villerusa@rsu.lv)
Lamberto Manzoli (lmanzoli@post.harvard.edu)

Version: 2 Date: 9 February 2010

Author’s response to reviews: see over
Torino, 9 February 2010

Dear Editor,

I'm happy to submit our revised methodological manuscript “A European study on alcohol and drug use among young drivers: the TEND by Night study design and methodology” for the publication on "BMC Public Health", after we ensured that our manuscript conformed to the journal style.

In particular the changes made in the manuscript were:

• a modification of capital letters in the title and in text subsections;
• a clarification of the ethical committee in the text;
• a section at the end of the manuscript listing the additional files;
• a mention of additional files in the text.

Moreover, the statement regarding ethics approval has been revised according to editorial request.

Looking forward to hearing from you,

Yours sincerely

Simone Chiadò Piat